Author’s response to reviews

Title: Multiple proteins and biological processes up-regulated during maxillary expansion in rats

Authors:

Junqing Ma (majunq@163.com)
Yunxia Wu (wuyunxia1126@126.com)
Weibing Zhang (zhangweibing@vip.sina.com)
Roger J Smales (roger.smales@adelaide.edu.au)
You Huang (huangyou1981@sina.com)
Yongchu Pan (panyongchu@tom.com)
Lin Wang (linwangnj@yahoo.com.cn)

Version: 4 Date: 11 March 2008

Author’s response to reviews:

Dear Editor,

Thank you very much for your letter dated 2008-3-10, regarding formatting requirements of our manuscript. We are grateful for your editorial efforts. We have now made the modifications suggested by you.

Thank you for your consideration. We are looking forward to publication of this manuscript.

Sincerely,
Dr Junqing Ma

Responses to Formatting Requirements:
1. Please ensure all text is black.
   [Response]:
   Done accordingly.

2. Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.
   [Response]:
   Based on this requirement, we have minimise white space around the image.

3. Website addresses- Websites must be removed from the text, and cited from the main reference list only.
   [Response]:
   Done accordingly.
4. References - References should be cited in the text using consecutive numbers in brackets. Please ensure that the order is correct.

[Response]:
Done accordingly.